Author's response to reviews

Title: A Polypill for primary prevention of cardiovascular disease: A feasibility study of the World Health Organization

Authors:

Elsayed Z Soliman (esoliman@wfubmc.edu)
Shanthi Mendis (mendiiss@who.int)
Wasantha P Dissanayake (wdissana@dialogsl.net)
Noel P Somasundaram (noelsomasundaram@gmail.com)
Padma S Gunaratne (pagunara@hotmail.com)
I Kumudini Jayasingne (kumujay@gmail.com)
Curt D Furberg (cfurberg@wfubmc.edu)

Version: 2 Date: 24 May 2010

Author's response to reviews:

We have revised the manuscript to include the names of the bodies that gave ethics approval